Author's response to reviews

Title: Improving person-centred care in nursing homes through dementia-care mapping: design of a cluster-randomised controlled trial.

Authors:

Geertje van de Ven (G.vandeVen@elg.umcn.nl)
Irena Draskovic (I.Draskovic@elg.umcn.nl)
Eddy MM Adang (E.Adang@ebh.umcn.nl)
Rogier ART Donders (R.Donders@ebh.umcn.nl)
Aukje Post (aukje.post@friesewouden.nl)
Syste U Zuidema (S.Zuidema@elg.umcn.nl)
Raymond TCM Koopmans (R.Koopmans@elg.umcn.nl)
Myrra JFJ Vernooij-Dassen (M.Vernooij-Dassen@iq.umcn.nl)

Version: 5 Date: 2 January 2012

Author's response to reviews: see over
Subject: Final revisions of the manuscript: *Improving person-centred care in nursing homes through dementia-care mapping: design of a cluster-randomised controlled trial.*

Dear Sir/Madam,

First we would like to thank you for providing us the opportunity to submit our manuscript entitled *Improving person-centred care in nursing homes through dementia-care mapping: design of a cluster-randomised controlled trial.* We submitted our paper by internet, with the changes to the format that were suggested in your email of 29 December 2011. We hope that by making these changes the manuscript conforms to journal style.
Yours faithfully, on behalf of the other authors,

Geertje van de Ven, MSc
Telephone: +31 (0)243655324
E-mail: G.vandeVen@elg.umcn.nl

Authors:
Geertje van de Ven¹, Irena Draskovic¹, Eddy MM Adang², Rogier ART Donders², Aukje Post³, Sytse U Zuidema¹, Raymond TCM Koopmans¹ and Myrra JFJ Vernooij-Dassen¹,4,5

¹ Eerstelijnsgeneeskunde, Radboud University Nijmegen Medical Centre (RUNMC), Nijmegen
² The Department of Epidemiology, Biostatistics and HTA, Radboud University Nijmegen Medical Centre (RUNMC), Nijmegen
³ Dementia Care Mapping, The Netherlands, De Friese Wouden, Burgum
⁴ Scientific Institute for Quality of Health Care, Radboud University Nijmegen Medical Centre (RUNMC), Nijmegen
⁵ Kalorama Foundation, The Netherlands, Beek-Ubbergen

Enclosures:
- Manuscript Improving person-centred care in nursing homes through dementia-care mapping: design of a cluster-randomised controlled trial.
- Additional file 1-3